Author's response to reviews

**Title:** A longitudinal qualitative study examining the factors impacting on the ability of persons with T1DM to assimilate the Dose Adjustment For Normal Eating (DAFNE) principles into daily living and how these factors change over time

**Authors:**

Dympna Casey (dympna.casey@nuigalway.ie)
Kathy Murphy (kathy.murphy@nuigalway.ie)
Julia Lawton (j.lawton@ed.ac.uk)
Florence Findlay-White (Florence.FindlayWhite@diabetes.org.uk)
Sean Dineen (sean.dineen@nuigalway.ie)

**Version:** 4  **Date:** 22 August 2011

**Author's response to reviews:**

The Structure of the manuscript has been reformatted and now has these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Authors' contributions; Authors¿ information (optional); Acknowledgements (optional); References